DiPALS: Diaphragm Pacing in patients with Amyotrophic Lateral Sclerosis – a randomised controlled trial

Christopher J McDermott, 1* Mike J Bradburn, 2 Chin Maguire, 2 Cindy L Cooper, 2 Wendy O Baird, 3 Susan K Baxter, 3 Judith Cohen, 2 Hannah Cantrill, 2 Simon Dixon, 4 Roger Ackroyd, 5 Simon Baudouin, 6 Andrew Bentley, 7 Richard Berrisford, 8 Stephen Bianchi, 9 Stephen C Bourke, 10 Roy Darlison, 11 John Ealing, 12 Mark Elliott, 13 Patrick Fitzgerald, 3 Simon Galloway, 7 Hisham Hamdalla, 12 C Oliver Hanemann, 14 Philip Hughes, 15 Ibrahim Imam, 16 Dayalan Karat, 17 Roger Leek, 18 Nick Maynard, 19 Richard W Orrell, 20 Abeezar Sarela, 13 John Stradling, 19 Kevin Talbot, 19 Lyn Taylor, 21 Martin Turner, 19 Anita K Simonds, 22 Tim Williams, 17 Wisia Wedzicha, 20 Carolyn Young 23 and Pamela J Shaw 1

¹Sheffield Institute for Translational Neuroscience, University of Sheffield, Sheffield, UK

²Sheffield Clinical Trials Research Unit, School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK

³School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK

⁴Health Economics and Decision Science, School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK

⁵Sheffield Teaching Hospitals NHS Foundation Trust, Northern General Hospital, Sheffield, UK

⁶Royal Victoria Hospital, Newcastle upon Tyne Hospitals NHS Foundation Trust and the University of Newcastle, Newcastle upon Tyne, UK

⁷University Hospital of South Manchester NHS Foundation Trust, Manchester, UK ⁸Plymouth Hospitals NHS Trust, Derriford Hospital, Plymouth, UK

⁹Sheffield Teaching Hospitals NHS Foundation Trust, Royal Hallamshire Hospital, Sheffield, UK

- ¹⁰North Tyneside General Hospital, Northumbria Healthcare NHS Foundation Trust and Newcastle University, North Shields, UK
- ¹¹Independent patient and public involvement representative, UK
- ¹²Salford Royal Hospitals NHS Foundation Trust, Salford, UK
- ¹³Leeds Teaching Hospitals NHS Trust, St James' University Hospital, Leeds, UK
- ¹⁴Plymouth University, Plymouth, UK
- ¹⁵Plymouth Hospitals NHS Trust Peninsula Medical and Dental Schools, Plymouth, UK
- ¹⁶South Devon Healthcare NHS Foundation Trust, Devon, UK
- ¹⁷Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle upon Tyne, UK
- ¹⁸Motor Neurone Disease Association, Birmingham, UK
- ¹⁹Oxford University Hospitals NHS Trust, Oxford, UK
- ²⁰The National Heart and Lung Institute, Imperial College London, London, UK
- ²¹PAREXEL International Corporation, Sheffield, UK
- ²²National Institute for Health Research Respiratory Biomedical Research Unit, Royal Brompton & Harefield NHS Foundation Trust, London, UK
- ²³Walton Centre for Neurology & Neurosurgery NHS Foundation Trust, Liverpool, UK

*Corresponding author

Declared competing interests of authors: Andrew Bentley reports honoraria and travel subsistence from BioMarin to attend clinical advisory group in Berlin (April 2015) on respiratory management of mucoploysaccharoidoses. Anita K Simonds reports being on the steering committee of trial of adaptive servo ventilation in heart failure patients with predominant obstructive sleep apnoea (SERVE-HF). Chin Maguire reports grants from Motor Neuron Disease Association and non-financial support from Synapse Biomedical Inc. during the conduct of the study. Christopher J McDermott reports grants from the National Institute for Health Research Health Technology Assessment programme, grants from the Motor Neurone Disease Association and non-financial support from Synapse Biomedical Inc. during the conduct of the study and separate to the Diaphragm Pacing in patients with Amyotrophic Lateral Sclerosis (DiPALS) trial. Lyn Taylor reports personal fees from PAREXEL International Corporation not associated with the DiPALS trial. Roger Leek reports membership of the Motor Neurone Disease Association. John Stradling reports personal fees from ResMed UK not associated with the DiPALS trial. Martin Turner reports grants from Medical Research Council, personal fees from Biogen Idec, personal fees from GlaxoSmithKline, personal fees from Neuraltus Pharmaceuticals, personal fees from Ontario Brain Institute, personal fees from GLG Consulting and personal fees from Vertex Pharmaceuticals not associated with the DiPALS trial. Richard W Orrell reports grants from the Motor Neurone Disease Association of England, Wales and Northern Ireland during the conduct of the study. Wisia Wedzicha reports grants from Johnson and Johnson, grants from Vifor Pharma, grants from Takeda, grants from GlaxoSmithKline, personal fees from Novartis, personal fees from GlaxoSmithKline, personal fees from Astra Zeneca, personal fees from Boehringer, personal fees from Takeda and personal fees from Vifor Pharma outside the submitted work.

Published June 2016 DOI: 10.3310/hta20450

Plain English summary

DiPALS

Health Technology Assessment 2016; Vol. 20: No. 45 DOI: 10.3310/hta20450

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

The research study aimed to compare two different treatments [non-invasive ventilation alone or non-invasive ventilation given alongside diaphragm pacing (DP)] for patients experiencing breathing difficulties due to amyotrophic lateral sclerosis (ALS). ALS is another name for motor neurone disease. Non-invasive ventilation is the treatment patients usually get where they wear a face mask and air is pushed into the lungs to help breathing. DP is a possible alternative treatment in which wires are inserted into the main breathing muscle to stimulate it directly to help patients breathe. The main outcomes of the study were to see if the addition of DP improved patient survival (how long patients lived) and improved quality of life.

We conducted our research in seven UK hospitals and enrolled 74 patients to the study (37 patients in each treatment group). Patients were enrolled between December 2011 and December 2013. The study stopped early based on recommendations from the committee charged with guarding patient safety. The committee felt that DP might be causing harm to patients. Patients in the group given DP alongside standard treatment had worse average survival (11.0 months) than those in the standard treatment only (non-invasive ventilation) group (22.5 months).

Based on study results, DP should not be used as routine treatment in ALS patients with breathing problems.

HTA/HTA TAR

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 5.027

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the ISI Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 09/55/33. The contractual start date was in July 2011. The draft report began editorial review in July 2015 and was accepted for publication in January 2016. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2016. This work was produced by McDermott *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Health Technology Assessment Editor-in-Chief

Professor Hywel Williams Director, HTA Programme, UK and Foundation Professor and Co-Director of the Centre of Evidence-Based Dermatology, University of Nottingham, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Professor Aileen Clarke Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Director of NETSCC, HTA, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Professor Elaine McColl Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Health Sciences Research, Health and Wellbeing Research and Development Group, University of Winchester, UK

Professor John Norrie Health Services Research Unit, University of Aberdeen, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk